Tutorial for week 7 (Feb 20, 2002)

February 15, 2002

The following questions will be considered during the tutorial sessions on Wednesday...

T5Q1: What is meant by “the MVC architecture” mentioned in section 5.3?

T5Q2: Investigate how you would create a “ToolTip” in Tcl/Tk - give a small code segment which demonstrates the ToolTip.

T5Q3: Investigate how you would create a “ToolTip” in Java/Swing - give a small code segment which demonstrates the ToolTip.

T5Q4: Write a minimal Java/Swing application which puts up a single File menu with a Quit item in it. T5Q5: The javax.swing.UIManager class is used to manipulate the look-and-feel of an application - as seen in section 5.3.1. How can you discover which look-and-feel strategies are implemented in the Java development environment?